Author’s response to reviews

Title: Association of parents’ alcohol use and family interaction with the initiation of alcohol use by sixth graders: A preliminary study in Taiwan

Authors:

Chao-Chia Hung (d93845004@ntu.edu.tw)
Lee-Lan Yen (leelan@ntu.edu.tw)
Wen-Chi Wu (wuvisan@ms17.hinet.net)

Version: 4 Date: 14 May 2009

Author’s response to reviews: see over
Dear Editor:

We are pleased to be advised that this paper has been accepted.

The name of ethical committee has been stated in the methods section on p.12.

Due to the article could be supported by a new grant from the “College of Public Health, National Taiwan University - Enhancing Capabilities for Social and Behavioral Sciences Research”, acknowledgement this institution and the grant in the manuscript is needed. We have supplemented the name of grant in the acknowledgement section on p.20.

Yours sincerely,

Lee-Lan Yen  
Sc. D Professor  
Institute of Health Policy and Management,  
College of Public Health,  
National Taiwan University.  
Room 623, No.17, Hsu-Chow Road, Taipei, Taiwan, R.O.C.  
Email: leelan@ntu.edu.tw